Author's response to reviews

Title: A Migraine Variant with Abdominal Colic and Alice in Wonderland Syndrome: A Case Report and Review

Authors:

    sherifa A Hamed (hamed_sherifa@yahoo.com)

Version: 4 Date: 19 November 2009

Author's response to reviews: see over
I appreciate the valuable the thoughtful reviewers’ comments which helped me to improve my manuscript to a great extent. The manuscript (MS: 2377696722882079) entitled “A Migraine Variant with Abdominal Colic and Alice in Wonderland Syndrome: A Case Report and Review” was revised taking into consideration all reviewers' comments (red color).

A DETAILED POINT-TO-POINT RESPONSE TO EACH REVIEWER'S COMMENTS:

Comments from the Associate Editor:
Get it copy-edited

Author:
- The whole manuscript was copy-edited by a colleague fluent in English language.

Comments from Referee 2:
The only correction that needs to be made is the sentence on the bottom of page 12 starting "Mutations of hereditary disorders may cause subtle alteration in the..." which doesn't make sense as written.

Author:
The sentence was corrected as follow:
“Changes in ion channels as a result of hereditary mutations could result in minimal or episodic changes in neuronal micro-environment. These changes might be sufficient to cause transient neuronal hyperexcitability.”

Comments from Referee 3:
Still far too long! I think worth 4 typed pages.

Author:
I shortened the manuscript and deleted some references, so it can fit 4-5 typed journal pages.

Again I appreciate the reviewer's valuable comments and advices.